

Received May 27, 2019, accepted June 30, 2019, date of publication July 22, 2019, date of current version August 6, 2019. *Digital Object Identifier* 10.1109/ACCESS.2019.2930225

Multimodal Neuroimaging Fusion in Nonsubsampled Shearlet Domain Using Location-Scale Distribution by Maximizing the High Frequency Subband Energy

EMIMAL JABASON[®], (Student Member, IEEE), M. OMAIR AHMAD[®], (Fellow, IEEE), AND M. N. S. SWAMY[®], (Fellow, IEEE)

Department of Electrical and Computer Engineering, Concordia University, Montreal, QC H3G 1M8, Canada Corresponding author: M. Omair Ahmad (omair@ece.concordia.ca)

This work was supported in part by the Natural Sciences and Engineering Research Council (NSERC) of Canada, and in part by the Regroupement Stratégique en Microsystèmes du Québec (ReSMiQ).

ABSTRACT A fusion of medical imaging data obtained from different modalities plays an important role in the current clinical practice. In this paper, we propose a novel multimodal fusion algorithm for brain imaging data based on the statistical properties of nonsubsampled shearlet transform (NSST) coefficients and a novel energy maximization fusion rule. The marginal distributions of the high-frequency NSST coefficients exhibit heavier tails than the Gaussian distribution. As a consequence, after studying its characteristics, we use a heavy-tailed probability density function, student's t location-scale distribution, to describe the highly non-Gaussian statistics of empirical NSST coefficients by learning the parameters using maximum likelihood estimation. Then, we employ this model to develop a maximum *a posteriori* estimator to obtain the noisefree coefficients. Then, for the first time, a novel fusion rule for obtaining the fused NSST coefficients based on maximizing the energy in the high-frequency subbands is proposed. Experiments are carried out on fusing two or more multimodal neuroimages taken from the BrainWeb, Alzheimer's Disease Neuroimaging Initiative (ADNI), and Whole Brain Atlas databases. It is seen from the subjective and objective results that the proposed multimodal neuroimaging fusion method significantly outperforms the state-of-the-art methods including under noisy scenarios, and hence, it is more robust. It is also observed that the signal intensities in the fused images are better enhanced when a more number of source images are being fused. The proposed technique should benefit the medical professionals in diagnosing neurological disorders, such as Alzheimer, epilepsy, and multiple sclerosis.

INDEX TERMS Maximum a posteriori estimation, multimodal fusion, neuroimaging data, nonsubsampled shearlet transform, student's *t* location-scale distribution.

I. INTRODUCTION

Brain is the most complex organ in the human body. Over the past 25 years, the burden of neurological disorders have increased significantly due to population growth and ageing [1]. In modern medicine, recent advances in noninvasive neuroimaging technology have had enormous impact on the diagnosis and treatment of brain diseases. In general, the neuroimaging technology falls into one of the two main categories, namely, structural and functional imaging. The structural neuroimaging deals only with the valuable anatomical information of the brain. As a result, the brain imaging data obtained from one of the above two techniques is not sufficient, in general, to carry out a proper diagnosis, and hence, data from multiple sources are acquired. As an example, structural magnetic resonance imaging (MRI) can detect changes in different regional structures such as lateral ventricle size with quite detailed resolution as shown in the Fig.1(a). On the other hand, the functional neuroimaging technique deals only with the functional neuroimaging technique deals only with the functional changes such as metabolic activity of neurons, local changes in blood flow, regional composition, and absorption of brain [2]. Moreover, it usually produces images of low resolution and fail to convey spatial information as shown in the Fig.1(b). In such situations, it is important to be able to see the relation between the anatomical details visible at one location in one modality

The associate editor coordinating the review of this manuscript and approving it for publication was Victor Hugo Albuquerque.



FIGURE 1. Difference in structural and functional imaging of a Alzheimer's disease patient from ADNI database. (a) MR image using magnetization-prepared rapid gradient-echo (MP-RAGE) (b) fludeoxyglucose (FDG) PET.

and the functional details that are more visible at the corresponding location in the other modality for better diagnosis and surgical planning. It is not easy to see such a relation by observing multiple modalities separately and to interpret and correlate the information provided by each modality. The desire to visualize such complementary information in a single image has led to investigations in multimodal image fusion. Such a fused image is more suitable for human and machine perception or for further image-processing tasks such as segmentation, feature extraction and object recognition. However, the existing fusion algorithms face a number of challenges such as loss of significant attributes, staircase effects, blocking artifacts, noisy patches, and contrast reduction. Specifically, our aim is to fuse 2D multimodal brain imaging data that results in a single fused image, which overcomes most of the existing challenges. It is useful to have a precise comparison of brain functional details with its underlying structure, avoiding possible mistakes resulting from alternate viewing of two or more separate images. However, the manual way of integrating the information obtained from multiple techniques is an extremely time-consuming and expensive process, which also demands years of experience to reduce human error. Hence, automated multimodal neuroimaging fusion is recognized as a promising solution to understand the different aspects of brain diseases at an early stage [3].

In recent years, several techniques have been developed to handle the multimodal fusion problem for neuroimaging data [3]–[20] based on spatial, or transform domain approaches [21]. Despite these efforts, it still remains a challenging task and requires more research to overcome some of the drawbacks such as staircase effects, loss of significant attributes, blocking artifacts, and contrast reduction of the existing automated fusion algorithms. In [6], the authors have developed a fusion method which is based on edgepreserving guided image filter. The idea is to first find two sharpened images by considering one as an input image and the other as a guidance image and vice versa. This is found by subtracting the input images from the blurred gradient information, it is clear that the intensity information could get lost. In addition, choosing a guidance image in case of fusing three or more images is a challenging task. Zhao and Lu [8] have proposed an adaptive fractional order total variation (TV) technique to estimate the fused image in the spatial domain. It can suppress the noise including the staircase effects present in the original TV models [22], [23]; but it still suffers from contrast reduction, just as other spatial domain techniques do [3]. In view of improved data representation, energy compaction, and reduced complexity, transform domain approaches are known to be more effective in handling multimodal image fusion problem [21]. The transform domain methods are generally categorized into sparse representation-based and multiscale decompositionbased methods. Liu et al. [11] have proposed a convolutional sparse representation-based fusion framework in which the source images are modeled as the sum over a set of convolutions between sparse coefficient maps and dictionary filters. In [9], the source image patches have been classified based on geometrical features and coded sparsely via online multiclass dictionary learning algorithm. The sparse representation-based methods learn dictionary elements from the source images. However, in case of noisy input, a highly redundant dictionary may cause unwanted visual artifacts in the reconstructed output [24]. Recently, multiscale transforms such as local Laplacian pyramid [5], wavelet [7], [25], curvelet [20], [21], contourlet [3], [19], shearlet [17], or cascade of such transforms [15], [26] have also been employed to develop fusion algorithms for neuroimaging and other medical images. In local Laplacian pyramid-based method [5], the source images are first decomposed using local Laplacian filtering [27] into approximate and residual images at different scales. Then, the approximate and residual images are fused using a local energy maximum scheme and an information of interest-based scheme, respectively. However, it is known that the widely-used Laplacian pyramid and wavelet transform is not effective in representing anisotropic features such as edges of multidimensional data. To overcome this limitation, Candès et al. [28] proposed the simpler, faster, and less redundant curvelet representation to capture the geometric information in images. In [20], a curvelet based fusion is implemented based on the application of the additive wavelet transform on the source images. The maximum fusion rule is applied on the ridgelet transform of high frequency bands. However, the implementation of curvelet in discrete domain is very challenging as it requires a rotation operation, which does not preserve the digital lattice [29]. To deal with this issue, two other multiresolution tools, contourlet [30] and shearlet [31] were introduced. Although these two transforms have good directional sensitivity, it is found that shearlet transform is more advantageous as there are no constraints on choosing the number of directions and the size of the supports for the shearing. It is also computationally more

images which is obtained using guided image filtering. Then, the sharpened images are combined based on the weights obtained through image statistics. Although it provides better

efficient unlike the usage of the directional filter banks in contourlet transform [31]. However, the downsamplers present in the Laplacian pyramid makes the shearlet transform, which is constructed by combining the Laplacian pyramid and shearing filters, shift invariant resulting in pseudo-Gibbs phenomena around the singularities such as edges [32]. Nonsubampled shearlet transform (NSST) [16], [31], [33] resolves this artifact by using nonsubampled filters in shearlet transform and it is proved as an excellent multiscale decomposition tool for medical image fusion [4], [34]. Taking this into account, in this paper, we employ NSST in decomposing the input data into different scales and directions. Furthermore, model-based fusion approaches [17], [25] have been developed using generalized Gaussian distribution (GGD) to characterize the non-Gaussian properties of subband coefficients. However, the traditional GGD model often exhibits exaggerated tails [35]. Hence, we proposed a statistical modeling technique [36] and found that student's t location-scale distribution has the potential to approximate the heavy tailed nature of NSST coefficients more closely when compared to GGD.

This paper introduces a novel Bayesian inference-based method for fusing neuroimaging data from an arbitrary number of modalities in the NSST domain using the student's t location-scale distribution as a prior for developing a posterior probability density function (PDF). In view of the fact that the high frequency subbands contain highly discriminative features such as edges, corners, blobs and ridges, a new rule for multimodal fusion based on maximizing the energy in all such subbands is proposed.

The paper is structured as follows. Section 2 reviews the basic structure of NSST. Section 3 describes the simplest forward model for neuroimaging fusion problem. Section 4 discusses the proposed neuroimaging fusion algorithm. Section 5 presents the experimental results of fusing two or more images using both synthetic data from BrainWeb [37], and real data from Alzheimer's disease neuroimaging initiative (ADNI) [38], and Whole Brain Atlas (WBA) [39] databases, followed by conclusion in Section 6.

II. NONSUBSAMPLED SHEARLET TRANSFORM

The nonsubsampled shearlet transform, a multiscale, multidirectional, and shift invariant framework proposed in [31], is found to be a highly efficient multiresolution tool to provide an optimal sparse approximation for multidimensional signals with anisotropic features such as edges and other spatial discontinuities present in images. In this section, we briefly discuss the implementation of NSST and the readers can find the complete details in [31]. In fact, shearlet is an expansion of composite wavelets introduced in [40]. Consider the 2-D affine system generated by $\upsilon \in L^2(\mathbb{R}^2)$, a collection of functions obtained by applying dilation and translation of the form:

$$\{v_{a,s,t}(m) = |\det G_{a,s}|^{\frac{1}{2}} \upsilon(G_{a,s}^{-1}m - t) :$$

$$a > 0, s \in \mathbb{R}, t \in \mathbb{R}^{2} \}$$
(1)

$$S\mathcal{H}_{\upsilon}: x \to S\mathcal{H}_{\upsilon}x(a, s, t) = \langle x, \upsilon_{a, s, t} \rangle$$
 (2)

The generating function v is a well-localized band-limited function and adequately meets the admissibility condition [41] associated with the shearlet approach so that $x \in L^2(\mathbb{R}^2)$ can be reconstructed by

$$x = \int_{\mathbb{R}^2} \int_{-\infty}^{\infty} \int_0^{\infty} \langle x, \upsilon_{a,s,t} \rangle \upsilon_{a,s,t} \, \frac{da}{a^3} \, ds \, dt \tag{3}$$

For any $\omega = (\omega_1, \omega_2) \in \mathbb{R}^2$, $\omega_1 \neq 0$, let Υ is chosen to be of the form:

$$\Upsilon(\omega) = \Upsilon_1(\omega_1)\Upsilon_2\left(\frac{\omega_2}{\omega_1}\right) \tag{4}$$

where Υ_1 and Υ_2 are smooth functions in which the supports are the proper subset of $[(-2, -\frac{1}{2})] \cup [\frac{1}{2}, 2]$ and [-1, 1], respectively. In the Fourier domain,

$$\Upsilon_{a,s,t}(\omega_1,\omega_2) = a^{\frac{3}{4}} \Upsilon_1(a\omega_1) \Upsilon_2\left(a^{-\frac{1}{2}}\left(\frac{\omega_2}{\omega_1} - s\right)\right) e^{-2\pi i\omega t}$$
(5)

where $\Upsilon_{a,s,t}(\omega_1, \omega_2)$ has frequency support in which ω_1 is in the set $[(-\frac{2}{a}, -\frac{1}{2a})] \cup [(\frac{2}{a}, \frac{1}{2a})]$ and $|\frac{\omega_2}{\omega_1} - s| \le \sqrt{a}$. Shearlet is a function of three parameters, namely, the scale *a*, the shear *s*, and, the translation *t*.

The discrete shearlet transform is obtained by discretizing the CST $S\mathcal{H}_{\nu}x(a, s, t) = \langle x, v_{a,s,t} \rangle$ on suitable discrete set (scaling, shear, and translation parameters) [42]. NSST is a special type of discrete shearlet transform and its decomposition is shown in Fig. 2. This figure depicts a two-scale (a = 2) decomposition, where the shear parameter s is chosen to be 8 and 4 at scales 1 and 2, respectively. The basic structure of NSST includes the nonsubsampled Laplacian pyramid (NSLP) transform, which establishes the multiscale property, in combination with shearing filters (SF) that offer a higher amount of directional information. Here, the use of nonsubsampled filters, without upsamplers or downsamplers, ensure the invariance property to the shift of the input functions. Given an image I of size $N \times N$, the procedure for finding the NSST at fixed scale l is summarized below [31].

- 1) Apply the NSLP to decompose *I* into a low-pass image
- L_j^l and a high-pass image H_j^l ; 2) Compute Pf_d^l , the discrete Fourier transform of H_j on the pseudo-polar grid;
- 3) Apply band-pass filtering to Pf_d^l ;
- 4) Apply inverse fast Fourier transform (IFFT) after re-assembling the samples on a Cartesian grid.



FIGURE 2. Block diagram of the NSST multiscale and multidirectional 2-scale decomposition. The nonsubsampled Laplacian pyramid (NSLP) is applied to the input function resulting in a low frequency subband, denoted by L, and a detailed subband, denoted by H. The detailed subband is fed into the shearing filter bank to obtain directional information.

As an example, the shearing filters (wedge shaped) computed using the Meyer wavelet function are shown in Fig. 3. The shearing filters of sizes 64×64 and 128×128 are used in the first and second scale of decomposition, respectively, with the number of shearing directions chosen to be 4 and 6 from small scale to large scale. The two-scale NSST decomposition of an MR image using maxflat filters in the NSLP stage is shown in Figs. 4 (b)-(1). Fig. 4 (b) is the low frequency sub-band, followed by the high frequency subbands at scale 1 and scale 2 in Figs. 4 (c)-(f) and Figs. 4 (g)-(1), respectively.

III. THE FORWARD LIKELIHOOD MODEL

Let $f_i(x, y)$ represent the unknown heterogeneous features such as structural and functional information of the brain, which are acquired by M different sensors and $g_i(x, y)$ the observed measurements. We consider neuroimaging fusion as the inverse of a linear forward model [22] that relates $f_i(x, y)$ and the observed brain images $g_i(x, y)$ by the following:

$$g_i(x, y) = f_i(x, y) + n_i(x, y)$$
 $i = 1, 2, ..., M$ (6)

where $n_i(x, y)$ is the noise associated with the *i*th sensor at location (x, y). In general, signal fluctuations originate in the physical processes of imaging rather than in the tissue textures. In MRI, the acquired real and imaginary data in the frequency domain are known to be corrupted by white noise having Gaussian probability distribution. The frequency domain data is transformed into Cartesian domain via inverse Fourier transform (IFT). After IFT, the real and imaginary parts are still corrupted by Gaussian noise, since the transform is linear and orthogonal. However, in reconstructed PET images, unlike MRI, the exact nature of the noise propagation is not well known. Hence, in this model, we assume that the target brain image is corrupted by additive white Gaussian noise with zero mean and a known standard deviation σ_n on each acquisition. Also, it is necessary that the input images are spatially aligned for developing image fusion algorithms [43]. Then, we apply non-subsampled shearlet transform (NSST) to all the observed images $g_i(x, y)$ to be fused. In general, if an image is decomposed into k scales



FIGURE 3. Examples of images of the shearing filters. Each detail subspace in NSST is chosen to be represented by four directions (a)-(d) in the first scale of decomposition and six directions (e)-(j) in the second scale of decomposition.



FIGURE 4. An example of two pyramidal scales nonsubsampled shearlet transform (NSST) decomposition. (a) MR image of an Alzheimer's disease patient. (b) The low frequency subband. (c)-(f) The high frequency subbands at scale 1. (g)-(l) The high frequency subbands at scale 2.



FIGURE 5. Block diagram of the proposed neuroimaging fusion method. Fusion takes place on individual channel for color images. STLS PDF stands for Student's t location-scale PDF.

and D(k) denotes the number of directions chosen at scale k, the NSST transform provides a low frequency subband at the finest scale and D(k) high frequency subbands at each scale k, in which the size of the subbands are similar to the size of the input image. We can write the model in (6) by an equivalent frequency domain representation:

$$G_i(u, v) = F_i(u, v) + N_i(u, v)$$
 $i = 1, 2, ..., M$ (7)

where

 $G_i(u, v) = \begin{cases} G_i^L(u, v) & \text{for the low frequency subband} \\ G_i^{k,d}(u, v) & \text{for the high frequency subbands} \end{cases}$

where

$$G_{i}^{L}(u, v) = F_{i}^{L}(u, v) + N_{i}^{L}(u, v)$$

$$G_{i}^{k,d}(u, v) = F_{i}^{k,d}(u, v) + N_{i}^{k,d}(u, v)$$

$$\begin{cases} i = 1, 2, \dots, M \\ k = 1, 2, \dots, K \\ d = 1, 2, \dots, D(k). \end{cases}$$

where the terms in the upper case letters are the NSSTs of the corresponding terms in (7) and M, K and D(k) denote the number of source images, number of scales and number of directions at each scale, respectively. The goal of the proposed neuroimaging fusion algorithm is to estimate the fused coefficients that combine the most significant information of $F_i^{k,d}(u, v)$ from the NSST coefficients of the given brain images, $g_i(x, y)$. Since we assume the probabilistic model associated with noisy NSST coefficients, $N_i^{k,d}(u, v)$, conditioned on $F_i^{k,d}(u, v)$ is Gaussian, the conditional probability is given by:

$$p(G_i(u, v)|F_i(u, v)) = p_{N_i(u, v)}(G_i(u, v) - F_i(u, v))$$

= $\mathcal{N}(0, \sigma_{N_i(u, v)}^2)$

which is proportional to

$$exp\frac{(G_i(u,v) - F_i(u,v))^2}{2\sigma_{N_i(u,v)}^2}$$
(8)



FIGURE 6. Empirical, generalized Gaussian and student's t location-scale CDFs for one of the NSST detail subbands of MR image of an AD patient.

We use a robust median estimator [44] to estimate the noise variance $\sigma_{N_i(\mu,\nu)}^2$) which is given by:

$$\sigma_{N_i(u,v)}^2 = MAD \frac{|Detail \ subbands \ of \ G_i(u,v)|}{0.6745}$$
(9)

where *MAD* is the median absolute deviation and $G_i(u, v)$ denotes the NSST coefficients of the observed neuroimaging data.

IV. PROPOSED NEUROIMAGING FUSION ALGORITHM

In this section, we compute the fused NSST coefficients statistically using Bayes' rule with some knowledge of the prior distribution. Fig.5 shows a schematic view of different modules involved in the proposed Bayesian inference-based multimodal neuroimaging fusion algorithm.



FIGURE 7. PP plots for empirical, student's t location-scale and GG distribution for four NSST detail subbands at the finest scale (a)-(d) for MR image, (e)-(h) for FDG-PET image and (i)-(l) for SPECT neuroimaging data.

A. STATISTICAL MODELING OF NSST COEFFICIENTS OF NEUROIMAGING DATA

At first, we propose to use the student's t location-scale PDF as a prior for modeling the heavy tailed nature of the NSST coefficients corresponding to the unknown target brain images under the assumption of independent identically distributed subbands [7]. Let x be a student's t location-scale distributed random variable; its PDF is described by

$$p(x|\nu,\mu_t,\sigma_t) = \frac{\Gamma(\frac{\nu+1}{2})}{\sigma_t \sqrt{\pi\nu} \ \Gamma(\frac{\nu}{2})} \left[\frac{\nu + (\frac{x-\mu_t}{\sigma_t})^2}{\nu} \right]$$
(10)

where $\sigma_t > 0$ is the scale parameter, μ_t is the location parameter and $\nu > 1$ is the shape parameter, which determine the nature of the distribution. The shape parameter, v is the most important parameter in controlling the shape of the distribution. When $\nu > 0$, the three-parameter student's t locationscale distribution behaves like a traditional one parameter student's t distribution. When $\nu \to \infty$, it approaches the Gaussian distribution. The smaller values of the shape parameter yields a sharp peak around zero. To validate the behavior of the proposed prior PDF qualitatively, we examine the cumulative distribution function of the empirical real NSST coefficients with the student's t location-scale and the generalized Gaussian (GG) PDF's of the neuroimaging data. We examine the statistical properties of NSST coefficients of different multimodal neuroimaging data. Fig. 6 emphasizes the modeling performance of the real NSST coefficients of MR neuroimaging data of an AD patient. In addition, Fig.7 shows the qualitative fitting results in terms of the P-P plot of empirical, student's t location-scale and GG for four NSST detail subbands of multimodal neuroimaging data such as MR, FDG-PET, and SPECT images. The above two figures show that the student's t location-scale distribution provides a better fit in comparison with that provided by the GG distribution.

1) PARAMETER ESTIMATION

In order to use student's t location scale distribution as a prior for estimating the fused coefficients, it is necessary to estimate the parameters from the NSST coefficients of the neuroimaging data. We employ the maximum likelihood estimation method [45] for learning the location, scale and shape parameters of the distribution. The maximum likelihood estimates are obtained through the expectation maximization (EM) algorithm [46], where the location and scale parameters can be estimated as

$$\mu_t^{(k+1)} = \frac{\sum_{i=1}^n W_i^{(k)} X_i}{\sum_{i=1}^n W_i^{(k)}}$$
(11)

$$\sigma_t^{(k+1)} = (1/n) \sum_{i=1}^n W_i^{(k)} (X_i - \mu_t^{(k)}) (X_i - \mu_t^{(k)})^T \quad (12)$$

where $W_i = \frac{\nu+1}{\nu+S_i}$ and $S_i = (X_i - \mu_t^{(k)})^T \sigma_t^{-1} (X_i - \mu_t^{(k)})$. Using the values of μ and σ , the following equation

$$\sum_{i=1}^{n} \left(-\psi\left(\frac{\nu^{(k+1)}}{2}\right) + \log\left(\frac{\nu^{(k+1)}}{2}\right) + 1 + R_i^{(k)} - r_i^{(k)} \right) = 0$$
(13)

with

$$r_{i}^{(k)} = \psi\left(\frac{\hat{\nu}^{(k)} + 1}{2}\right) - \log\left(\frac{1}{2}\left(\nu^{(k)} + (x_{i} - \hat{\mu}_{t}^{(k)})^{T} \times \hat{\sum}^{(k)^{-1}}(x_{i} - \hat{\mu}_{t}^{(k)})\right)\right)$$
(14)

can be solved numerically to estimate the shape parameter. In summary, the EM algorithm for the maximum likelihood estimator is as follows: Given the estimates at iteration, k, the $k + 1^{th}$ iteration of EM consists of the following two steps: expectation (E) step and maximization (M) step. The E step computes the expected value of the log likelihood, given the observed data and current estimate of the model parameters. The M step maximizes the resulting function with respect to the model parameters, yielding new estimates to be used in the next iteration. These steps are repeated until convergence criterion is satisfied. Furthermore, for quantitative analysis, we use the Jensen-Shannon divergence (JSD) [47] test to calculate the goodness of fit between the empirical and fitted distributions. The JSD metric, a symmetric form of Kullback-Leibler divergence (KLD), is a measure of similarity between two probability distributions and is defined by

$$D_{KL}(p(x), q(x)) = \sum_{x} p(x) \ln \frac{p(x)}{q(x)}$$
(15)

$$D_{KL}(q(x), p(x)) = \sum_{x} q(x) \ln \frac{q(x)}{p(x)}$$
(16)

$$JSD = \frac{D_{KL}(p(x), q(x)) + D_{KL}(q(x), p(x))}{2}$$
(17)

where p(x) is the PDF of the fitted random variable and q(x) is the PDF of the empirical data. The JSD values corresponding to the student's t location-scale and GG distributions

Sut	bonde	N	/IR	FDC	G-PET	SP	ЕСТ
Bui	manus	GGD	STLSD*	GGD	STLSD*	GGD	STLSD*
	1	0.0045	0.0020	0.0345	0.0021	0.0223	0.0028
	2	0.0049	0.0029	0.0228	0.0022	0.0342	0.0023
I	3	0.0045	0.0015	0.0045	0.0023	0.0089	0.0031
	4	0.0058	0.0023	0.0082	0.0081	0.0065	0.0034
	5	0.0068	0.0031	0.0098	0.0090	0.0089	0.0067
	6	0.0023	0.0019	0.0073	0.0073	0.0045	0.0034
	1	0.0065	0.0021	0.0213	0.0032	0.0185	0.0027
	2	0.0025	0.0021	0.0066	0.0058	0.0065	0.0024
	3	0.0048	0.0024	0.0088	0.0076	0.0056	0.0045
	4	0.0047	0.0023	0.0405	0.0193	0.0304	0.0032
п	5	0.0043	0.0021	0.0078	0.0076	0.0256	0.0087
	6	0.0030	0.0012	0.0074	0.0034	0.0089	0.0045
	7	0.0045	0.0019	0.0078	0.0065	0.0065	0.0034
	8	0.0029	0.0013	0.0105	0.0067	0.0045	0.0023
	9	0.0065	0.0025	0.0387	0.0145	0.0082	0.0076
	10	0.0034	0.0015	0.0278	0.0256	0.0067	0.0056
	11	0.0045	0.0022	0.0069	0.0025	0.0077	0.0024

 TABLE 1. Goodness-of-fit measures in terms of jensen-shannon divergence.

* Student's t location-scale distribution

in Fig. 6 are 0.0030 and 0.0078, respectively. Table 1 shows the comparison between GG and student's t location-scale distributions by averaging the values of JSD metric computed over various subbands for different neuroimaging data. It can be seen that the student's t location-scale fits the empirical cumulative distribution function better than the popularlyused GG distribution does.

B. FUSION DECISION RULE

In Section IV(A), we have shown the modeling ability of student's t location-scale distribution for the NSST coefficients of neuroimaging data. Therefore, after estimating the parameters, this well approximated distribution is used as a prior PDF to estimate the noise-free coefficients using MAP estimator in which we find $F_i^{k,d}(u, v)$ that maximizes the posterior PDF $p(\frac{F_i(u,v)}{G_i(u,v)})$. Since the approximation coefficients, $F_i^L(u, v)$ deal with low frequency information, i.e., the coarse representation of the image, we follow averaging operation to combine the coefficients of the low frequency subbands of the various images [48]. The detailed coefficients contain high frequency information including important details such as edges and corners present in the images in different directions.

As the coefficients of the high frequency subbands are complex numbers, one of the most commonly used procedures for finding the decision map is the absolute maximum selection rule [49]. According to this rule, when two images are fused, the fused coefficient $F_f^{k,d}(u, v)$ is $F_1^{k,d}(u, v)$ or $F_2^{k,d}(u, v)$ depending on whether $|F_1^{k,d}(u, v)| \ge |F_2^{k,d}(u, v)|$ or $|F_1^{k,d}(u, v) < |F_2^{k,d}(u, v)|$, where $F_1^{k,d}(u, v)$ and $F_2^{k,d}(u, v)$

are the coefficients located at (u, v) in the subband at the k^{th} scale and in the d^{th} direction of the images 1 and 2, respectively. Thus, in the absolute maximum selection rule, the energy of the fused image at the position (u, v) is that of image 1 or image 2 at that position depending on the one that is larger. Thus, using this rule, the energy level of the entire fused image is, in general, larger than that of either of the two images used for fusion. The improvement in the quality of the fused image over that of the individual images can be attributed to the increased level of its energy. In order to further improve the quality of the fused image, one should be investigating an approach that will increase the energy level at its individual pixel position even further. For this purpose, we now propose the following fusion rule.

we now propose the following fusion rule. Let $F_i^{k,d}(u, v) = F_{iR}^{k,d}(u, v) + jF_{il}^{k,d}(u, v)$ be the transform domain coefficient located at (u, v) in the high frequency subband at the k^{th} scale and in the d^{th} direction of the image i = 1, 2, 3, ..., M and $F_f^{k,d}(u, v) = F_{fR}^{k,d}(u, v) + jF_{fI}^{k,d}(u, v)$ be the corresponding coefficient at (u, v) in the subband at the k^{th} scale and in the d^{th} direction of the fused image. The fusion rule is defined as

$$F_{fR}^{k,d}(u,v) = \sum_{i} W_{iR}(u,v) F_{iR}^{k,d}(u,v)$$
(18a)

and

$$F_{fI}^{k,d}(u,v) = \sum_{i} W_{iI}(u,v) F_{iI}^{k,d}(u,v)$$
(18b)

where the decision maps, $W_{iR}(u, v)$ and $W_{il}(u, v)$ are defined by (19), as given at the bottom of this page.

It is to be mentioned that if there is more than one value of *i* that satisfies $i = \mu$ (or $i = \gamma$), then we choose any one of the *i* values as μ (or γ) and let the corresponding weight to '1', and the rest of the weights to '0'.

Thus, in the proposed fusion rule, the real and imaginary parts of the fused coefficient $F_f^{k,d}(u, v)$ in a given high frequency subband are obtained separately, as opposed to that in the absolute maximum selection rule. The real (imaginary) part of the fused coefficient $F_{fR}^{k,d}(u, v) \left(F_{fI}^{k,d}(u, v)\right)$ is chosen to be the one for which its absolute value is the largest amongst the absolute values of the real (imaginary) parts of $F_i^{k,d}(u, v)$ of the various images involved in the fusion.

It can easily be seen that, in general,

$$|F_f^{k,d}(u,v)| > |F_i^{k,d}(u,v)|$$
(20)

for all values of *i*. Thus, the total energy in the high frequency subbands using the proposed fusion rule will always be greater than that obtained using absolute maximum

$$W_{iR}(u,v) = \begin{cases} 1, & i = \mu, & \text{if } |F_{\mu R}^{k,d}(u,v)| \ge |F_{1R}^{k,d}(u,v)|, |F_{2R}^{k,d}(u,v)| \dots |F_{MR}^{k,d}(u,v)| \\ 0, & i \ne \mu \end{cases}$$
(19a)

$$W_{iI}(u,v) = \begin{cases} 1, & i = \gamma, & \text{if } |F_{\gamma I}^{k,d}(u,v)| \ge |F_{1I}^{k,d}(u,v)|, |F_{2I}^{k,d}(u,v)| \dots |F_{MI}^{k,d}(u,v)| \\ 0, & i \ne \gamma \end{cases}$$
(19b)



FIGURE 8. Fusion results using the conventional and proposed fusion rules. (a) and (b) are the pair of source images being fused. (c) and (d) are the fused images using conventional and proposed fusion rules, respectively.

selection rule. We will refer to the proposed fusion rule as the energy maximizing fusion rule.

C. IMPLEMENTATION STEPS

In this subsection, we summarize the main steps involved in the proposed medical image fusion algorithm.

- 1) Apply the NSST transform on the individual source image.
- 2) Estimate the parameters μ_t , σ_t , ν from the NSST coefficients of the each input image using the method described in Section IV(A).
- Apply the fusion rule on the NSST coefficients as explained in Section IV(B).
- 4) Apply the inverse NSST to the fused coefficients obtained in Step 3.

V. EXPERIMENTAL RESULTS AND DISCUSSION

In this section, we study the performance of the proposed technique on different pairs of 2-D multimodal synthetic and real brain images. Also, we compare the performance of the proposed method with that of five other methods, namely, curvelet transform-based (CTB) method [20], contourlet transform-based (CB) method [19], guided image filter-based (GIFB) method [6], local Laplacian filtering domain-based (LLDB) method [5], and parameter-adaptive pulse coupled neural network in nonsubsampled shearlet transform-based (PA-PCNN-NSSTB) method [4]. For a fair comparison, we choose a number of pairs of images from various databases (BrainWeb [37], Alzheimer's disease neuroimaging initiative (ADNI) [38], and Whole Brain Atlas (WBA) [39]), and obtain the fusion results for each of the methods mentioned above as well as for the proposed method. For the methods of [5], [6], we have used the codes published by the authors to obtain the fusion results, whereas for the other methods [4], [19], [20], we have written codes in Matlab. For the purpose of obtaining the experimental results, we select two scales for the contourlet and non-subsampled shearlet transforms to decompose the source images, where for the first scale six directions and for the second scale ten directions are chosen. Thus, in total, we have 16 high frequency subbands and one low frequency subband. For quantitative analysis, five performance metrics, peak signal to noise ratio (PSNR), structural similarity (SSIM), mutual information (MI), entropy and $Q^{\frac{f_1f_2}{F}}$ are used, and these are briefly defined next. The SSIM measure is a function of three independent components, namely, luminance, contrast, and structure.

$$SSIM(x, y) = [l(x, y)]^{\alpha} [c(x, y)]^{\beta} [s(x, y)]^{\gamma}$$
(21)

where α , β , and γ are the quantities used to calibrate the relative importance of these particular components. We choose $\alpha = \beta = \gamma = 1$; the readers are referred to [50] concerning the details on the choice of the parameters and the steps involved in the calculation of the SSIM metric. In the context of image fusion, the MI metric measures the amount of information the fused image conveys about each of the source images. Considering two input images $f_1(x, y)$ and $f_2(x, y)$ and a fused image F(x, y), we can calculate how much information F(x, y) holds from $f_1(x, y)$ and $f_2(x, y)$ based on Kullback-Leibler measure in (11)

$$I(F, f_k) = \sum_{i=1}^{L} \sum_{j=1}^{L} p_{Ff_k}(i, j) \log \frac{p_{Ff_k}(i, j)}{p_F(i)p_{f_k}(j)} k = 1, 2$$
(22)

where $p_{Ff_k}(i, j)$ is the joint histogram of images F(x, y) and $f_k(x, y)$, p_F and p_{f_k} are the normalized histograms of F(x, y) and $f_k(x, y)$, respectively, and L is the number of intensity values. Thus, the image fusion MI measure is defined by simply taking the average of the composite image F(x, y) and each of the input images $f_1(x, y)$, and $f_2(x, y)$, i.e.,

$$MI_F^{f_1f_2} = \frac{I(F, f_1) + I(F, f_2)}{2}$$
(23)

Higher the value of MI better the quality of the fused image. Another metric that is used to measure the quality of fusion is normalized weighted performance metric $Q^{\frac{f_1f_2}{F}}$. It calculates the amount of edge information transfers from each of the input images into the fused image. The normalized weighted performance measure $Q^{\frac{f_1f_2}{F}}$ of a given fused image *F* that operates on input images f_1 and f_2 is defined as follows: $Q^{\frac{f_1f_2}{F}}$

$$= \frac{\sum_{x=1}^{M} \sum_{y=1}^{N} Q^{f_1 F}(x, y) w^{f_1}(x, y) + Q^{f_2 F}(x, y) w^{f_2}(x, y)}{\sum_{i=1}^{M} \sum_{j=1}^{N} w^{f_1}(i, j) + w^{f_2}(i, j)}$$
(24)

where $w^{f_1}(m, n)$ and $w^{f_2}(m, n)$ are the edge strength values of the input images f_1 and f_2 , respectively, at each pixel location (x,y) and are used as weight factors in (24). $Q^{f_1F}(x, y)$ and $Q^{f_2F}(x, y)$ are the edge information preservation values of f_1 and f_2 , respectively, in the fused image F at each pixel location (x, y). Each of these quantities is the product of the edge strength preservation values and the orientation preservation values calculated by using the Sobel operator [51], [52]. The value of the normalized weighted performance measure $Q^{\frac{f_1f_2}{F}}$ ranges between 0 (complete loss of edge information) and 1 (no loss of edge information) [52]. As explained in [17], for color images, we first calculate the performance measures SSIM, MI, and $Q^{\frac{f_1f_2}{F}}$ of an individual channel separately using (21), (23) and (25), respectively, followed by averaging out over the number of channels. As an example, for color images (f)

(f)

IEEE Access

(h)

(h)



(e)

FIGURE 9. Multimodal fusion results of 2-D simulated neuroimages of a normal brain taken from the brainWeb database. (a) and (b) correspond to MR-T1 and MR-PD images of a normal brain, respectively. (c), (d), (e), (f), (g), and (h) correspond to the fused images using curvelet-based [20], contourlet-based [19], guided image filter-based [6], local Laplacian filtering domain-based [5], parameter-adaptive pulse coupled neural network and nonsubsampled shearlet-based [4], and proposed methods, respectively.

(g)



FIGURE 10. Multimodal fusion results of 2-D simulated neuroimages of a multiple sclerosis lesion brain taken from the brainWeb database. (a) and (b) correspond to MR-T2 and MR-PD images of a multiple sclerosis lesion brain, respectively. (c), (d), (e), (f), (g), and (h) correspond to the fused images using curvelet-based [20], contourlet-based [19], guided image filter-based [6], local Laplacian filtering domain-based [5], parameter-adaptive pulse coupled neural network and nonsubsampled shearlet-based [4], and proposed methods, respectively.

(g)

(e)



FIGURE 11. Multimodal fusion results of 2-D simulated neuroimages with 40% intensity non-uniformity level and 5% noise taken from the brainWeb database. (a) MR-T1 image. (b) MR-PD image. (c), (d), (e), (f), (g), and (h) correspond to the fused images using curvelet-based [20], contourlet-based [19], guided image filter-based [6], local Laplacian filtering domain-based [5], parameter-adaptive pulse coupled neural network and nonsubsampled shearlet-based [4], and proposed methods, respectively.

such as RGB, we calculate MI by considering the red, green and blue color components of each individual pixel so that (23) becomes

$$MI_F^{f_1f_2} = \frac{1}{3} \sum_{i}^{R,G,B} \frac{I^i(F,f_1) + I^i(F,f_2)}{2}$$
(25)

The other common performance measures, PSNR and entropy, are also calculated in the same manner for the color images.

We first investigate the energy values obtained in each high frequency subband when the absolute maximum selection rule and the proposed energy maximizing fusion rule are applied. For this purpose, we consider Figs. 8 (a) and (b) taken from the WBA database. It is found that the energy level of each subband is increased using the proposed fusion rule, with a total increase of 7.15% over that using the absolute maximum selection rule. We now compare the performance of our method that involves student's t location scale modeling using the proposed fusion rule with that using the conventional absolute maximum selection rule. The objective quality of the fused images, using the two rules, as measured by the various metrics, is summarized in Table 2. It is seen from this table that there is an improvement in the values of every one of the metrics indicating the performance using the proposed fusion rule is superior to that using the absolute maximum selection rule. This improved performance is further reinforced by the fused images shown in Figs. 8 (c) and (d) using the absolute maximum selection and proposed fusion

TABLE 2. Comparison of conventional and proposed fusion rules in terms of objective fusion measures.

Source Image	Fusion Rule	PSNR	SSIM	MI	Entropy	$Q^{\frac{f_1f_2}{F}}$
Fig. 8 (a) and (b)	Absolute Maximum Selection	16.0222	0.5946	1.5009	4.2751	0.7327
	Proposed	17.1243	0.6334	1.5705	5.7234	0.8507

rules, respectively. This improvement is clearly due to the use of the proposed fusion rule, which retains important high frequency details such as edges, corners, blobs and ridges better than the conventional absolute maximum selection rule does. Hence, in all our subsequent experiments, fusion is carried out using the novel energy maximizing fusion rule.

A. EXPERIMENTAL RESULTS USING SIMULATED DATA

To assess the performance of our proposed method, we carry out extensive experiments on pairs of simulated brain MR images taken from the BrainWeb database. For this purpose, we consider slices along the axial plane from the longitudinal relaxation (T1)-weighted, transverse relaxation (T2)-weighted, and proton density (PD)-weighted MR brain volume with various slice thicknesses, intensity nonuniformity levels, and noise. Fig. 9 illustrates the results of fusion of T1-weighted MR (MR-T1) and PD-weighted MR (MR-PD) images of a normal brain. The objective quality of the fused images obtained for the various methods as measured by the different metrics is summarized in Table 3. It is seen from this table that the proposed method yields the highest values in all the measures considered. This is further reinforced by observing the visual quality of the fused images

Source Images	Method	PSNR	SSIM	МІ	Entropy	$Q^{rac{f_1f_2}{F}}$
	CTB [20]	16.0849	0.552	2.2389	5.398	0.683
Fig 0	CB [19]	16.2785	0.6292	2.2553	5.0648	0.7204
(a) and (b)	GIFB [6]	15.8754	0.6735	2.2145	4.4678	0.6689
	LLDB [5]	16.9352	0.7346	2.3267	5.3237	0.7864
	PA_PCN_NNSSTB [4]	17.0287	0.7425	2.2261	5.8458	0.8009
	Proposed	17.2185	0.7948	2.5338	6.5722	0.8397
	CTB [20]	16.9616	0.728	2.653	5.4947	0.8869
Fig 10	CB [19]	17.1494	0.8042	2.7742	5.1992	0.9023
(a) and (b)	GIFB [6]	16.9869	0.8465	2.6167	4.8734	0.8367
(a) and (b)	LLDB [5]	17.2165	0.8498	2.6854	5.5841	0.8592
	PA_PCN_NNSSTB [4]	17.3674	0.8357	2.3268	5.5845	0.8612
	Proposed	17.5287	0.8525	2.8461	5.7358	0.9309

shown in Figs.9 (c)-(h) using the curvelet-based, contourletbased, guided image filter-based, local Laplacian filtering domain-based, parameter-adaptive pulse coupled neural network in nonsubsampled shearlet transform-based, and proposed methods, respectively. Fig. 10 illustrates the results of fusion of T2-weighted MR (MR-T2) and PD-weighted MR (MR-PD) images of a multiple sclerosis brain. The objective quality of the fused images obtained for the various methods as measured by the metrics is also summarized in Table 3. It is also seen from this table that the proposed method yields the highest values in all the measures considered. This is further reinforced by observing the visual quality of the fused images shown in Figs.10 (c)-(h) using the curvelet-based, contourletbased, guided image filter-based, local Laplacian filtering domain-based, parameter-adaptive pulse coupled neural network in nonsubsampled shearlet transform-based, and proposed methods, respectively. From these two figures, it is clear that the proposed method is more efficient in transferring the structural details from the MR images of T1/T2 and PD weighting techniques compared to that obtained using other traditional methods.

In addition, we now consider the same two MR images shown in Figs. 9 (a) and (b), but with 40% intensity nonuniformity level and containing 1% or 5% or 9% noise. The noise in the simulated scans has Rayleigh statistics in the background and Rician statistics in the signal regions [37]. The fusion results as measured by the various metrics is summarized in Table 4. It is seen from this table that the proposed method has the highest values in all of the five performance metrics considered. This is further reinforced by the subjective fusion results shown in Fig. 11, when the noise is 5%. It is observed from the fused images that even in the noisy case, unlike the other traditional methods, the proposed method could retain most of the structural information of the source images in the fused image, and provide a better image quality. Thus, the proposed method is more robust than the other traditional methods.

B. EXPERIMENTAL RESULTS USING REAL DATA

In this subsection, we carry out extensive experiments on th fusion of real neuroimaging data taken from Alzheimer's Disease Neuroimaging Initiative (ADNI) [38] and Whole Brain Atlas (WBA) [39] databases.

Noisy Data	Method	PSNR	SSIM	MI	Entropy	$Q^{\frac{f_1f_2}{F}}$
	CTB [20]	20.1979	0.7005	2.4777	5.4376	0.7285
40% intensity	CB [19]	20.3785	0.7489	2.5907	5.2155	0.7521
non-uniformity level	GIFB [6]	19.9827	0.7139	2.5125	5.1003	0.7067
and	LLDB [5]	19.8325	0.7540	2.6322	5.4289	0.7932
9% noise	PA-PCNN-NSSTB [4]	19.3576	0.7653	2.5632	5.4162	0.7843
	Proposed	20.4189	0.781	2.6795	5.5379	0.819
	CTB [20]	17.9937	0.6042	2.1455	5.254	0.6672
40% intensity	CB [19]	18.1873	0.674	2.1728	4.9356	0.7009
non-uniformity level	GIFB [6]	17.7914	0.6489	2.0689	4.7135	0.6589
and	LLDB [5]	17.9842	0.7043	2.284	5.2165	0.7621
5% noise	PA-PCNN-NSSTB [4]	18.654	0.7216	2.3412	5.1685	0.7634
	Proposed	18.733	0.7315	2.5779	5.4988	0.8157
	CTB [20]	16.672	0.5713	2.193	5.3405	0.675
40% intensity	CB [19]	16.8476	0.6473	2.2212	5.0154	0.711
non-uniformity level	GIFB [6]	16.4853	0.6279	2.1098	4.7365	0.6664
and	LLDB [5]	16.5731	0.687	2.2491	5.3727	0.7210
1% noise	PA-PCNN-NSSTB [4]	17.7462	0.6932	2.1560	5.1356	0.8127
	Proposed	17.7885	0.7076	2.3567	5.5424	0.8396

TABLE 4. Objective fusion results of MR images with noise.

TABLE 5. Objective fusion results of 2-D neuroimages in Fig. 12.

Source Images	Method	PSNR	SSIM	MI	Entropy	$Q^{rac{f_1f_2}{F}}$
	CTB [20]	20.2011	0.7022	1.4183	5.2607	0.5756
Fig 12	CB [19]	21.2807	0.8776	2.5294	4.9487	0.5842
(a) and (b)	GIFB [6]	21.1421	0.8456	2.3876	4.6322	0.7920
(a) and (b)	LLDB [5]	21.4567	0.8823	2.4762	5.9876	0.6237
	PA-PCNN-NSSTB [4]	21.4938	0.8923	2.3921	5.6854	0.7932
	Proposed	21.5599	0.8977	2.8484	6.9367	0.8567

1) FUSION OF TWO MR IMAGES TAKEN FROM ADNI DATABASE

We now consider the fusion of a pair of real MR images taken from the Alzheimer's disease neuroimaging initiative (ADNI) database. The images in Fig. 12 (a) and (b) are the slices along the axial plane from the MR-T1 and MR-PD volumes, respectively. Table 5 lists the objective fusion results for various methods. It is observed from this table that the proposed method outperforms all the other methods in terms of all the measures considered. In particular, it has a significantly higher entropy as well as a normalized weighted performance metric, $Q^{\frac{f_1f_2}{F}}$. This is further reinforced by observing the visual quality of the fused images shown in Figs. (c)-(h) using the curvelet-based, contourlet-based, guided image filter-based, local Laplacian filtering domainbased, parameter-adaptive pulse coupled neural network in nonsubsampled shearlet transform-based, and proposed methods, respectively. It is seen from these fused images that the proposed technique preserves the structural details from both the T2 and PD weighting techniques better than the other traditional methods do.

2) FUSION OF TWO MR IMAGES TAKEN FROM WBA DATABASE

We now consider the fusion of all possible pairs of MR-T1, MR-T2, and MR-PD images of a lyme encephalopathy patient and a mild Alzheimer's disease patient taken from the Whole Brain Atlas (WBA) database. Table 6 lists the objective performance measures for the three pairs of 2-D MR images of the lyme encephalopathy patient, while Table 7 lists the corresponding results for the mild Alzheimer's disease



FIGURE 12. Multimodal fusion results of 2-D real neuroimages of an Alzheimer's disease patient from the ADNI database. (a) MR-T2 image. (b) MR-PD image. (c), (d), (e), (f), and (g) correspond to the fused images using curvelet-based [20], contourlet-based [19], guided image filter-based [6], local Laplacian filtering domain-based [5], parameter-adaptive pulse coupled neural network and nonsubsampled shearlet-based [4], and proposed methods, respectively.

TABLE 6. Objective fusion results of 2-D neuroimages in Fig. 13.

Source Images	Method	PSNR	SSIM	МІ	Entropy	$Q^{\frac{f_1f_2}{F}}$
	CTB [20]	15.1990	0.6660	1.8584	4.6795	0.5841
Eia 12	CB [19]	16.2012	0.5455	1.6728	5.2058	0.6289
(a) and (b)	GIFB [6]	16.2961	0.6582	1.7470	5.2219	0.6678
(a) and (b)	LLDB [5]	14.1159	0.5800	1.5545	4.8573	0.6579
	PA_PCN_NNSSTB [4]	16.8567	0.6371	2.0316	5.1834	0.6531
	Proposed	19.7274	0.7031	2.1381	5.2479	0.7299
	CTB [20]	17.8508	0.6719	1.8183	4.6789	0.5748
Ela 12	CB [19]	17.9119	0.5912	1.6636	5.5128	0.6558
Fig.15	GIFB [6]	18.0121	0.6812	1.7471	4.9856	0.6943
(a) and (c)	LLDB [5]	15.1264	0.5530	1.3744	3.8218	0.5250
	PA-PCNN-NSSTB [4]	16.2853	0.6398	1.9384	5.0508	0.5013
	Proposed	21.6344	0.7927	2.2404	5.1395	0.7386
	CTB [20]	17.9729	0.6822	1.8206	4.5653	0.5678
Ela 12	CB [19]	18.0572	0.5868	1.7059	5.5985	0.6545
rig.15	GIFB [6]	18.1884	0.6830	1.7776	5.0653	0.6798
(\mathbf{D}) and (\mathbf{C})	LLDB [5]	15.5506	0.5732	1.4375	4.9315	0.6059
	PA-PCNN-NSSTB [4]	16.7662	0.6427	1.9194	5.0926	0.5331
	Proposed	20.8525	0.8508	2.4986	5.1180	0.7204

patient. It is seen from these tables that the proposed technique outperforms all the other methods in terms of all the measures considered. This is further reinforced by the subjective fusion results shown in Fig. 13 for the lyme encephalopathy patient and the corresponding results shown in Fig.14 for the mild Alzheimer's disease patient. It is seen from these fused images that the proposed technique preserves the structural details of the source images for all the three pairs better than the other traditional methods do.

3) FUSION OF AN MR IMAGE WITH SPECT OR PET IMAGE TAKEN FROM WBA DATABASE

We now consider the fusion of four pairs of multimodal brain images taken from the Whole Brain Atlas (WBA) database.

TABLE 7. Objective fusion results of 2-D neuroimages in Fig.14.

Source Images	Method	PSNR	SSIM	MI	Entropy	$Q^{\frac{f_1f_2}{F}}$
	CTB [20]	17.4250	0.6996	1.9849	4.9650	0.7186
Fig 14	CB [19]	16.4331	0.5997	1.8998	4.7807	0.7484
(a) and (b)	GIFB [6]	16.6803	0.7106	2.0190	4.2428	0.7681
	LLDB [5]	15.6440	0.4904	1.5521	4.5880	0.5905
	PA_PCN_NNSSTB [4]	16.8253	0.7061	1.9313	4.9968	0.7577
	Proposed	18.5626	0.8047	2.0743	5.3081	0.8357
	CTB [20]	15.5317	0.8137	2.4057	4.8595	0.8507
Fig 14	CB [19]	15.7643	0.7484	2.3738	5.0257	0.8652
Fig.14	GIFB [6]	14.1377	0.8354	2.4763	5.0616	0.8741
(a) and (c)	LLDB [5]	15.6020	0.5483	1.6156	4.4413	0.6012
	PA-PCNN-NSSTB [4]	17.8390	0.8093	2.5994	5.3449	0.7612
	Proposed	18.2550	0.8929	2.6848	5.6313	0.8017
	CTB [20]	16.8453	0.8324	2.4018	4.9674	0.8467
Fig 14	CB [19]	16.8374	0.7725	2.2875	5.1198	0.8808
(b) and (a)	GIFB [6]	17.0589	0.8417	2.3992	5.1825	0.8905
(b) and (c)	LLDB [5]	15.3579	0.5174	1.4647	4.6997	0.5197
	PA-PCNN-NSSTB [4]	17.3163	0.8191	2.5185	5.0034	0.8191
	Proposed	18.9036	0.8500	2.6662	5.2467	0.8155

The various pairs of images are shown in Fig. 15. The pair of images in Figs. 15(a) and (e) correspond to T2-weighted magnetic resonance imaging (MR-T2) and positron emission tomography using F-18-fluorodeoxy-glucose (FDG-PET) from a 70 year-old man with mild Alzheimer's disease, respectively. The pair of images in Figs. 15(b) and (f) correspond to MR-T2 and single photon emission computed tomography with Thallium-201 (SPECT-T1) from a 51 year old woman with Anaplastic Astrocytoma, a type of rare malignant tumor, respectively. The pair of images in Figs. 15(c) and (g) are that of the T1-weighted MR (MR-T1), after gadolinium-diethylenetriamine pentaacetic acid (Gd-DTPA)and FDG-PET from a 53 year old man with Astrocytoma, a type of cancer of the brain, respectively.

IEEEAccess



FIGURE 13. Multimodal fusion results of 2-D real neuroimages of a lyme encephalopathy patient from the whole brain atlas database. (a) MRI-T1 image. (b) MR-T2 image. (c) MR-PD image. Rows 2 to 4 correspond to the fusion images of the input combinations (a) and (b), (a) and (c), and (b) and (c), respectively. Here, columns 1 to 6 correspond to the fused images of curvelet-based [20], contourlet-based [19], guided image filter-based [6], local Laplacian filtering domain-based [5], parameter-adaptive pulse coupled neural network and nonsubsampled shearlet-based [4], and proposed methods, respectively.

Finally, the pair of images in Figs. 15 (d) and (h) are MR-T2 and SPECT with perfusion agent Tc99m-HM-PAO (SPECT-Tc) from a 76 year old man with Subdural Hygroma, accumulation of cerebrospinal fluid in the subdural membrane. The objective performance measures obtained for each of the four pairs of 2-D neuroimages using various fusion methods are summarized in Table 8. It is seen from this table that the proposed method consistently outperforms all the other methods in terms of all the objective measures considered and in particular, it has significantly higher values of the normalized weighted performance metric, $Q^{\frac{f_1 f_2}{F}}$, signifying that the proposed fusion algorithm conveys the gradient information without any significant loss.

The fused images for each of the four pairs using the curveletbased method are shown in Figs.15 (i)-(l), the corresponding fused images using the contourlet-based method are shown in Figs.15 (m)-(p), those using the guided image filter-based are shown in Figs.15 (q)-(t), those using the local Laplacian filtering domain-based are shown in Figs.15(u)-(x), and those using the parameter-adaptive pulse coupled neural network in nonsubsampled shearlet transform-based are shown in Figs.15 (y)-(ab). The fused images using the proposed method for each of the four pairs are shown in Figs.15 (ac)-(af). In order to appreciate the effectiveness of the proposed method over that of the other methods in preserving both the structural and functional information of



FIGURE 14. Multimodal fusion results of 2-D real neuroimages of a mild Alzheimer's disease patient from the whole brain atlas database. (a) MR-T1 image. (b) MR-T2 image. (c) MR-PD image. Rows 2 to 4 correspond to the fusion images of the input combinations (a) and (b), (a) and (c), and (b) and (c), respectively. Here, columns 1 to 6 correspond to the fused images of curvelet-based [20], contourlet-based [19], guided image filter-based [6], local Laplacian filtering domain-based [5], parameter-adaptive pulse coupled neural network and nonsubsampled shearlet-based [4], and proposed methods, respectively.

the two modalities of each pair of images, a small segment (red box as shown in Figs.15 (a)-(h)) of each pair of the original images and the corresponding segments of the fused images obtained by various methods are all zoomed in and the resulting zoomed segments are shown in Fig. 16. It is clearly seen from Fig.16 that the proposed technique preserves the structural features such as the edge information in MR and the functional features such as the color information in both PET and SPECT images better than the other existing techniques do. This is in view of the fact the NSST used in our approach provides optimal representation of neuroimaging data with edges and texture, and further, the student's t location-scale distribution, which acts a prior distribution, closely fits the empirical NSST coefficients of source images; more importantly, it is due to the proposed fusion rule, which increases the overall high frequency subband energy. Also, it is noticed that the guided image filter scheme produces blocking artifacts reflecting the lack of sufficient directional information. Although the local Laplacian filtering scheme provides good contrast and no blocking artifacts, the functional information is not well preserved as observed from the color details in Fig.16 (v) and some noise is introduced as seen from Fig.16 (u). Also, it is noticed from the fused image, shown in Fig. 16(ad), that substantial structural information is lost in lieu of functional superimposition. However, only a subset of the large region of a mixed signal on MR-T2 corresponds

IEEE Access

						X	
(a)	(b)	(c)	(d)	(a)	(b)	(c)	(d)
			E		8		
(e)	(f)	(g)	(h)	(e)	(f)	(g)	(h)
					6	A	
(i)	(j)	(k)	(1)	(i)	(j)	(k)	(1)
(m)	(n)	(0)	(p)	(m)	(n)	(0)	(p)
(q)	(r)	(s)	(t)	(q)	(r)	(s)	(t)
					6		
(u)	(v)	(w)	(x)	(u)	(v)	(w)	(x)
(1)	(7)	(22)	(ab)	(y)	(z)	(aa)	(ab)
(y)	(2)	(aa)	(ab)	FIGURE 16. Multin shown in Fig.14 (a) four images in each in Fig.14. to the active tun	nodal fusion result - (h)] of all the pa h row correspond nor which is dif	ts of the zoom reginits of neuroimage to the enlarged are fficult to localiz	on [red box as s in Fig.12. The ea of the images ee. The thallium
(ac)	(ad)	(ae)	(af)	uptake of tumor	region in SPE	CT has been sh	own to increase
(ac)	(au)	(ac)	(ar)	the sensitivity a Hence, fusing	and specificity these two imag	of tumor detec ges provides cl	tion over MRI. inicians with a

FIGURE 15. Multimodal fusion results of 2-D real neuroimages from the whole brain atlas database. (a) and (e), (b) and (f), (c) and (g), and (d) and (h) are the pairs of source images being fused. Rows 3 to 8 correspond to the fusion results of curvelet-based [20], contourlet-based [19], guided image filter-based [6], local Laplacian filtering domain-based [5], parameter-adaptive pulse coupled neural network and nonsubsampled shearlet-based [4], and proposed methods, respectively.

VOLUME 7, 2019

97879

single image that combines higher sensitivity of the color

SPECT image with the improved localization of the suspected

weighted performance metric, $Q^{\frac{f_1f_2}{F}}$ obtained using the

Fig. 17 shows the bar graph of the average normalized

details in the MRI image.



FIGURE 17. Objective multimodal fusion results comparison in terms of $Q^{\frac{f_1 f_2}{F}}$.

TABLE 8.	Objective	fusion	results of	2-D	neuroimag	es in	Fig.15	5.
----------	-----------	--------	------------	-----	-----------	-------	--------	----

Source Images	Method	PSNR	SSIM	МІ	Entropy	$Q^{\frac{f_1f_2}{F}}$
	CTB [20]	15.6906	0.544	1.4931	5.1231	0.5949
Fig 15	CB [19]	15.6585	0.5358	1.4557	5.1337	0.6118
(a) and (a)	GIFB [6]	15.3278	0.534	1.465	5.5401	0.5934
	LLDB [5]	15.9276	0.5281	1.4972	5.6984	0.6038
	PA_PCN_NNSSTB [4]	16.7136	0.5982	1.4987 5.5872	0.7912	
	Proposed	16.7246	0.6334	1.5705	5.7234	0.8507
	CTB [20]	16.9026	0.5997	1.3604	4.8770	0.6077
Fig 15	CB [19]	16.8106	0.5795	1.2915	5.0118	0.6442
(b) and (f)	GIFB [6]	16.1564	0.5876	1.1332	5.3483	0.6745
	LLDB [5]	16.9254	0.6237	1.392	5.4872	0.7238
	PA-PCNN-NSSTB [4]	16.3749	0.6357	1.2968	5.4845	0.7612
	Proposed	16.9308	0.6573	1.4086	5.6932	0.8525
	CTB [20]	19.1321	0.6348	1.9821	5.4808	0.6033
Fig 15	CB [19]	19.1351	0.6233	1.9478	5.543	0.6372
(a) and (a)	GIFB [6]	18.3678	0.6342	1.7654	6.1425	0.6859
(C) and (g)	LLDB [5]	19.5923	0.6821	1.9943	6.1893	0.7815
	PA-PCNN-NSSTB [4]	20.0674	0.6324	1.9269	5.6956	0.7874
	Proposed	20.2543	0.6926	2.0739	6.2802	0.8351
	CTB [20]	20.326	0.5904	2.1641	6.1583	0.6251
Fig 15	CB [19]	20.2649	0.5834	2.1123	6.1593	0.6302
(d) and (b)	GIFB [6]	20.1453	0.5673	2.1432	6.5690	0.6003
	LLDB [5]	20.9246	0.56958	2.2871	6.9326	0.7152
	PA-PCNN-NSSTB [4]	20.9434	0.6053	2.1468	5.9865	0.7362
	Proposed	21.5452	0.6547	2.3827	7.0571	0.8096

proposed and other traditional methods for the multimodal pairs, namely, MR-T1/SPECT, MR-T2/SPECT, MR-PD/SPECT, MR-T1/PET, MR-T2/PET, and MR-PD/PET. It is seen from this figure that the proposed technique outperforms the other traditional methods in terms of the normalized weighted performance metric, $Q_{\frac{f_1f_2}{F}}^{f_1f_2}$ for all the multimodal pairs considered. Ideally, the $Q_{\frac{f_1f_2}{F}}^{f_1f_2}$ values of 0 and 1 indicate the loss of and complete recovery of edge information, respectively.

We next consider the fusion of an MR image and SPECT image of a lyme encephalopathy patient taken from the WBA database.

4) FUSION OF AN MR IMAGE WITH THE SPECT IMAGE OF A LYME ENCEPHALOPATHY PATIENT

We now consider fusing each of the MR images of a lyme encephalopathy patient shown in Figs. 18 (a), (b), and (c) with the functional SPECT image shown in Fig. 18 (d). Figs.18(a), (b), and (c) are, respectively, the same as Figs. 13 (a), (b), and (c), and are repeated here for convenience. Table 9 lists the corresponding objective fusion results. The corresponding subjective fusion results are shown in Fig. 18. It is clear from these results that the proposed method performs better than the traditional methods do in terms of both the objective and subjective fusion results.

We next consider the fusion of an MR image and FDG-PET image of a mild Alzheimer's disease patient taken from the WBA database.



FIGURE 18. Multimodal fusion results of an MR image with the SPECT image of a lyme encephalopathy patient. (a) MR-T1 image (b) MR-T2 image (c) MR-PD image, and (d) SPECT image. (a) and (d), (b) and (d), and (c) and (d) are the pairs of source images being fused. (e)-(j), (k)-(p), and (q)-(v) are the fused images of (a) and (d), (b) and (d), and (c) and (q), respectively. Rows 3 to 8 correspond to the fused images obtained using curvelet-based [20], contourlet-based [19], guided image filter-based [6], local Laplacian filtering domain-based [5], parameter-adaptive pulse coupled neural network and nonsubsampled shearlet-based [4], and proposed methods, respectively. NOTE: Figs. 18(a), (b), and (c) are, respectively, the same as Figs. 13 (a), (b), and (c), and are repeated here for convenience.



FIGURE 19. Multimodal fusion results of an MR image with the PET image of a mild Alzheimer's disease patient. (a) MR-T1 image (b) MR-T2 image (c) MR-PD image, and (d) FDG-PET image. (a) and (d), (b) and (d), and (c) and (d) are the pairs of source images being fused. (e)-(j), (k)-(p), and (q)-(v) are the fused images of (a) and (d), (b) and (d), and (c) and (d), respectively. Rows 3 to 8 correspond to the fused images obtained using curvelet-based [20], contourlet-based [19], guided image filter-based [6], local Laplacian filtering domain-based [5], parameter-adaptive pulse coupled neural network and nonsubsampled shearlet-based [4], and proposed methods, respectively. NOTE: Figs. 19 (a), (b), and (c) are, respectively, the same as Figs. 14 (a), (b), and (c), and are repeated here for convenience.

TABLE 9. Objective fusion results of 2-D neuroimages in Fig.18.

Source Images	Method	PSNR	SSIM	MI	Entropy	$Q^{\frac{f_1f_2}{F}}$
	CTB [20]	15.8341	0.5929	2.3998	5.7207	0.6484
Eig 19	CB [19]	15.5331	0.5947	2.8998	5.7817	0.6484
(a) and (d)	GIFB [6]	16.6803	0.5136	2.0190	5.2438	0.6681
(a) and (d)	LLDB [5]	16.5470	0.5904	2.5521	5.5840	0.7905
	PA_PCN_NNSSTB [4]	15.8253	0.6261	2.9313	6.9968	0.6977
	Proposed	20.5426	0.6947	2.8743	7.3781	0.8357
	CTB [20]	17.5348	0.6359	1.8456	4.6535	0.5968
Eig 19	CB [19]	16.5120	0.5472	1.6932	5.3938	0.7328
(b) and (d)	GIFB [6]	18.1522	0.6902	1.7491	4.9052	0.6693
(b) and (d)	LLDB [5]	15.7244	0.5801	1.4794	3.5617	0.6307
	PA-PCNN-NSSTB [4]	16.9849	0.6532	2.1038	5.1954	0.5013
	Proposed	18.6264	0.6976	2.0968	5.9629	0.7855
	CTB [20]	17.9678	0.6127	1.8702	4.8238	0.5376
Eig 18	CB [19]	18.2474	0.6438	1.7162	5.1034	0.6257
(a) and (d)	GIFB [6]	18.2906	0.6746	1.5689	5.1263	0.6985
(c) and (u)	LLDB [5]	15.5468	0.5762	1.5638	4.9336	0.6278
	PA-PCNN-NSSTB [4]	16.4669	0.6367	1.9267	5.0689	0.5731
	Proposed	19.9033	0.6992	2.8728	5.1568	0.7025

TABLE 10. Objective fusion results of 2-D neuroimages in Fig.19.

Source Images	Method	PSNR	SSIM	MI	Entropy	$Q^{\frac{f_1f_2}{F}}$
	CTB [20]	15.6707	0.6369	1.8564	4.6118	0.5530
Eta 10	CB [19]	15.6985	0.5274	1.6706	5.6544	0.6286
Fig.19	GIFB [6]	15.7968	0.6325	1.7519	5.1185	0.6515
(a) and (d)	LLDB [5]	14.3914	0.5737	1.4910	3.3657	0.6262
	PA_PCN_NNSSTB [4]	15.6722	0.5850	1.8903	4.9748	0.5102
	Proposed	16.1480	0.7141	1.8925	5.7751	0.8255
Fig.19 (b) and (d)	CTB [20]	15.6616	0.6258	1.7394	4.2633	0.5484
	CB [19]	15.6993	0.5164	1.5765	5.2628	0.6013
	GIFB [6]	15.7978	0.6185	1.6470	4.6844	0.6423
	LLDB [5]	14.9290	0.5774	1.2805	3.8800	0.5908
	PA-PCNN-NSSTB [4]	15.6880	0.5816	1.7446	4.6772	0.5015
	Proposed	16.3369	0.6313	1.8431	5.5082	0.8180
Fig.19 (c) and (d)	CTB [20]	17.4610	0.6582	1.8526	4.3291	0.5391
	CB [19]	17.4849	0.5683	1.7143	5.3242	0.6264
	GIFB [6]	17.5935	0.6585	1.8073	4.7825	0.6565
	LLDB [5]	16.8937	0.5726	1.4906	4.6048	0.5070
	PA-PCNN-NSSTB [4]	16.7705	0.5976	1.8744	4.7313	0.4751
	Proposed	17.7378	0.7553	1.8656	5.6774	0.7874

TABLE 11. Objective fusion results of three 2-D MR images.

Source Images	PSNR	SSIM	MI	Entropy	$Q^{\frac{f_1 f_2 f_3}{F}}$
Fig. 20 (a),(b), and (c)	22.8285	0.9878	2.7578	5.5354	0.7417
Fig. 21 (a),(b), and (c)	19.1549	0.8948	2.8636	5.6831	0.8863

5) FUSION OF AN MR IMAGE WITH THE FDG-PET IMAGE OF A MILD ALZHEIMER'S DISEASE PATIENT

We now consider fusing each of the MR images of a mild Alzheimer's disease patient shown in Figs. 19 (a), (b), and (c) with the functional FDG-PET data shown in Fig. 19 (d). Figs. 19(a), (b), and (c) are, respectively, the same as Figs. 14 (a), (b), and (c), and are repeated here for convenience. Table 10 lists the corresponding objective fusion results. The corresponding subjective fusion results are shown in Fig. 19. It is clear from these results that the proposed method performs better than the traditional methods do in terms of both the objective and subjective fusion results.

6) EXPERIMENTS ON FUSING MORE THAN TWO IMAGES TAKEN FROM WBA DATABASE

We now consider the fusion of three 2-D brain images from the Whole Brain Atlas database (WBA) using the proposed method. Since data is not available for considering the fusion



FIGURE 20. Multimodal fusion of three 2-D real MR neuroimages of a lyme encephalopathy patient from the whole brain atlas database. (a) MR-T1 image. (b) MR-T2 image. (c) MR-PD image. (d) Fused image of (a), (b) and (c).



FIGURE 21. Multimodal fusion of three 2-D real MR neuroimages of a mild Alzheimer's disease patient from the whole brain atlas database. (a) MR-T1 image. (b) MR-T2 image. (c) MR-PD image. (d) Fused image of (a), (b) and (c).

of more than two functional images, we have considered only the fusion of three MR images that reveal structural information.

To compare the performance of the proposed method on fusing three images with that of fusing two images, we choose the source images, MR-T1, MR-T2, and MR-PD shown in Fig.13 (a), (b), and (c), respectively, of the lyme encephalopathy patient. The objective quality of the fused image as measured by the various metrics is summarized in Table 11. By comparing these results with that shown in Table 6, it is observed that the values of the various metrics when all the three images are fused are higher than those when any two of the three images are fused. This is further confirmed by the subjective fusion results. Fig. 20 shows the fusion results of the three multimodal MR images. It is clearly seen from Fig. 20 (d) that the signal intensities are well enhanced when all the different weighting techniques, namely, T1, T2, and PD of MR images are considered for fusion unlike fusing only any two different weighting techniques of MR images as in Fig. 13. Similar experiments are carried out for the source images, MR-T1, MR-T2, and MR-PD shown in Fig.14 (a), (b), and (c), respectively, of the mild Alzheimer's disease patient. The corresponding objective fusion results are given in Table 11 and the subjective fusion results are shown in Fig.21.

We next consider the fusion of two MR images and one PET/SPECT image. To compare the performance of the proposed method on fusing three images with that of

TABLE 12.	Objective fusion results of three images (two MR images and
one PET/SI	PECT image).

Source Images	PSNR	SSIM	MI	Entropy	$Q^{\frac{f_1 f_2 f_3}{F}}$
Fig. 22 (a),(b), and (d)	23.1210	0.9629	2.9646	5.9635	0.8894
Fig. 22 (a),(c), and (d)	22.2547	0.8607	2.9695	5.5574	0.8400
Fig. 22 (b),(c), and (d)	21.9836	0.8119	2.8986	5.4101	0.8460
Fig. 23 (a),(b), and (d)	19.6532	0.8874	1.8538	5.6973	0.8953
Fig. 23 (a),(c), and (d)	19.3368	0.8448	1.4480	5.5478	0.8966
Fig. 23 (b),(c), and (d)	19.9019	0.8542	1.3277	5.2921	0.8892

fusing two images, we choose the source images, MR-T1, MR-T2, MR-PD, and SPECT shown in Fig.18 (a), (b), (c), and (d), respectively, of the lyme encephalopathy patient. The combinations are MR-T1/MR-T2/SPECT, MR-T1/ MR-PD/SPECT, MR-T2/MR-PD/SPECT. The objective quality of the fused images obtained using as measured by the various metrics is summarized in Table 12. By comparing these results with that shown in Table 9, it is observed that the values of four of the five metrics considered are higher than those when any one of the MR images and the SPECT image are fused at a time. The subjective fusion results are shown in Fig. 22. It is clear from the fused images that the features are well enhanced when any two of the MR images and the SPECT image are fused unlike fusing any one of the MR images and SPECT image. Similar experiments are carried out for the source images, MR-T1, MR-T2, MR-PD, and PET images shown in Fig. 19 (a), (b), (c), and (d) respectively, of the mild Alzheimer's disease patient. The corresponding objective fusion results are given in Table 12. By comparing

IEEEAccess



FIGURE 22. Multimodal fusion of three 2-D real neuroimages (MR-SPECT combinations) of a lyme encephalopathy patient from the whole brain atlas database. (a) MR-T1 image. (b) MR-T2 image. (c) MR-PD image. (d) SPECT image. (e) Fused image of (a), (b), and (d). (f) Fused image of (a), (c), and (d). (g) Fused image of (b), (c), and (d). NOTE: Figs. 22 (a), (b), and (c) are, respectively, the same as Figs. 18 (a), (b), and (c), and are repeated here for convenience.





FIGURE 23. Multimodal fusion of three 2-D real neuroimages (MR-PET combinations) of a mild Alzheimer's disease patient from the whole brain atlas database. (a) MR-T1 image. (b) MR-T2 image. (c) MR-PD image. (d) FDG-PET image. (e) Fused image of (a), (b), and (d). (f) Fused image of (a), (c), and (d). (g) Fused image of (b), (c), and (d). NOTE: Figs. 23 (a), (b), and (c) are, respectively, the same as Figs. 19 (a), (b), and (c), and are repeated here for convenience.

these results with that shown in Tables 10 and 12, it is observed that the proposed method yields the highest values in all the measures considered except the mutual information and entropy measures, when any two of the MR images and the PET image are fused. The subjective fusion results are shown in Fig.23. It is clear from the fused images that the

TABLE 13. Objective fusion results of four images (three MR images and one PET/SPECT image).

Source Images	PSNR	SSIM	MI	Entropy	$Q^{\frac{f_1 f_2 f_3 f_4}{F}}$
Fig. 22 (a),(b),(c), and (d)	23.5487	0.9783	2.8745	5.1237	0.8981
Fig. 23 (a),(b),(c), and (d)	20.1342	0.8983	1.8755	5.6332	0.8997



FIGURE 24. Multimodal fusion of four 2-D real neuroimages (MR-PET combinations) from the whole brain atlas database. (a) Fused image of Figs. 22 (a), (b), (c), and (d). (b) Fused image of Figs. 23 (a), (b), (c), and (d).

features are well enhanced when any two of the MR images and the PET image are fused unlike fusing any one of the MR images and PET image.

Finally, we consider the fusion of four input images, MR-T1, MR-T2, MR-PD, and SPECT, shown in Figs. 22 (a), (b), (c), and (d), respectively, of the lyme encephalopathy patient. The objective quality of the fused image obtained using as measured by the various metrics is summarized in Table 13. By comparing these results with that shown in Tables 9 and 12, it is observed that the proposed method yields the highest values in all the measures considered except the mutual information and entropy and measures, when any one or two of the MR images and the SPECT image are fused. We believe that these scores could be further increased by considering more number of functional images. We also notice that the proposed technique boosts the normalized weighted performance metric. $O^{\frac{f_1 f_2}{F}}$ as the number of images to be fused increases. Fig. 24 (a) is the subjective fusion result. It is clear from the fused image that when we integrate the structural features from different MR images along with the functional data, the features are well enhanced unlike having only two or three images. Similar experiments are carried out for the input images, MR-T1, MR-T2, MR-PD, and PET, shown in Figs. 23 (a), (b), (c), and (d), respectively, of the mild Alzheimer's disease patient. The corresponding objective fusion results are given in Table 13 and the subjective fusion result is shown in Fig. 24 (b).

As mentioned earlier, in our experiments we have chosen two scales, one with six and the other with ten directions. However, the performance could be further improved by increasing the number of scales and directions at the expense of increased complexity. In future, a study could be undertaken to evaluate the trade-off between the fusion performance and the computational complexity resulting from different numbers of scales and directions.

VI. CONCLUSION

In this paper, for the first time using the concept of energy maximization in high frequency subbands, a novel method for multimodal neuroimaging fusion has been proposed. The proposed method has employed the NSST transforms of the images whose coefficients are modeled using student's t location-scale distribution and fused using the novel energy maximizing fusion rule. It has been shown that this distribution results in a better approximation of the empirical NSST coefficients of the neuroimaging data than the traditional generalized Gaussian distribution does. A Bayesian MAP estimator has been devised by using this distribution as a prior in order to obtain the noise-free coefficients of the images. The new fusion rule is based on the idea of maximizing the total energy of the high frequency subbands in the fused image. This has been achieved by separately fusing the real and imaginary parts of the corresponding high frequency coefficients of the images being fused.

Experiments have been carried out to assess the performance of the proposed method. Based on our results using both synthetic and real brain data, it has been shown that the proposed method significantly outperforms the other traditional methods in terms of all the performance measures considered, namely, peak signal to noise ratio (PSNR), structural similarity (SSIM), mutual information (MI), entropy and $O^{\frac{f_1f_2}{F}}$. In addition, from the subjective results, it is clear that the proposed method preserves the structural and functional details better than the other existing techniques do. It is also evident that the traditional challenges such as blocking artifacts, noisy patches, and loss of significant attributes have been overcome and hence, the proposed method is more robust, thus producing fused images of higher quality. The proposed method is completely general in that it can be applied to any number of neuroimages, subject to their availability. It has been found that the signal intensities in the fused images are better enhanced when more number of source images are being fused.

The improved performance of the proposed method is due to the fact that the NSST transform captures the image details in different scales and directions, and the student's t locationscale PDF, which acts as a prior for Bayesian inference-based fusion, approximates the NSST coefficients very well, and more importantly in view of the maximization of energy in the high frequency subbands using the proposed fusion rule. We believe that the framework developed here for multimodal neuroimaging fusion will help in the study of brain diseases, such as Alzheimer's, epilepsy, and multiple sclerosis.

REFERENCES

- V. L. Feigin *et al.*, "Global, regional, and national burden of neurological disorders during 1990–2015: A systematic analysis for the global burden of disease study 2015," *Lancet Neurology*, vol. 16, no. 11, pp. 877–897, Nov. 2017.
- [2] J. Woo, M. Stone, and J. L. Prince, "Multimodal registration via mutual information incorporating geometric and spatial context," *IEEE Trans. Image Process.*, vol. 24, no. 2, pp. 757–769, Feb. 2015.

- [3] S. Das and M. K. Kundu, "A neuro-fuzzy approach for medical image fusion," *IEEE Trans. Biomed. Eng.*, vol. 60, no. 12, pp. 3347–3353, Dec. 2013.
- [4] M. Yin, X. Liu, Y. Liu, and X. Chen, "Medical image fusion with parameter-adaptive pulse coupled neural network in nonsubsampled shearlet transform domain," *IEEE Trans. Instrum. Meas.*, vol. 68, no. 1, pp. 49–64, Jan. 2019.
- [5] J. Du, W. Li, and B. Xiao, "Anatomical-functional image fusion by information of interest in local laplacian filtering domain," *IEEE Trans. Image Process.*, vol. 26, no. 12, pp. 5855–5866, Dec. 2017.
- [6] D. P. Bavirisetti, V. Kollu, X. Gang, and R. Dhuli, "Fusion of MRI and CT images using guided image filter and image statistics," *Int. J. Imag. Syst. Technol.*, vol. 27, no. 3, pp. 227–237, Sep. 2017.
- [7] S. Liu, "Study on medical image enhancement based on wavelet transform fusion algorithm," *J. Med. Imag. Health Informat.*, vol. 7, no. 2, pp. 388–392, Apr. 2017.
- [8] W. Zhao and H. Lu, "Medical image fusion and denoising with alternating sequential filter and adaptive fractional order total variation," *IEEE Trans. Instrum. Meas.*, vol. 66, no. 9, pp. 2283–2294, Sep. 2017.
- [9] J.-J. Zong and T.-S. Qiu, "Medical image fusion based on sparse representation of classified image patches," *Biomed. Signal Process. Control*, vol. 34, pp. 195–205, Apr. 2017.
- [10] V. Calhoun and J. Sui, "Multimodal fusion of brain imaging data: A key to finding the missing link(s) in complex mental illness," *Biol. Psychiatry, Cognit. Neurosci. Neuroimag.*, vol. 1, no. 3, pp. 230–244, May 2016.
- [11] Y. Liu, X. Chen, R. K. Ward, and Z. J. Wang, "Image fusion with convolutional sparse representation," *IEEE Signal Process. Lett.*, vol. 23, no. 12, pp. 1882–1886, Dec. 2016.
- [12] M. Manchandaa and R. Sharma, "A novel method of multimodal medical image fusion using fuzzy transform," J. Vis. Commun. Image Represent., vol. 40, pp. 197–217, Oct. 2016.
- [13] J. Du, W. Li, B. Xiao, and Q. Nawaz, "Union laplacian pyramid with multiple features for medical image fusion," *Neurocomputing*, vol. 194, pp. 326–339, Jun. 2016.
- [14] J. Du, W. Li, K. Lu, and B. Xiao, "An overview of multi-modal medical image fusion," *Neurocomputing*, vol. 215, pp. 3–20, Nov. 2016.
- [15] V. Bhateja, H. Patel, A. Krishn, A. Sahu, and A. Lay-Ekuakille, "Multimodal medical image sensor fusion framework using cascade of wavelet and contourlet transform domains," *IEEE Sensors J.*, vol. 15, no. 12, pp. 6783–6790, Dec. 2015.
- [16] S. Singh, D. Gupta, R. S. Anand, and V. Kumar, "Nonsubsampled shearlet based CT and MR medical image fusion using biologically inspired spiking neural network," *Biomed. Signal Process. Control*, vol. 18, pp. 91–101, Apr. 2015.
- [17] L. Wang, B. Li, and L. Tian, "Multimodal medical volumetric data fusion using 3-D discrete shearlet transform and global-to-local rule," *IEEE Trans. Biomed. Eng.*, vol. 61, no. 1, pp. 197–206, Jan. 2014.
- [18] R. Shen, I. Cheng, and A. Basu, "Cross-scale coefficient selection for volumetric medical image fusion," *IEEE Trans. Biomed. Eng.*, vol. 60, no. 4, pp. 1069–1079, Apr. 2013.
- [19] G. Bhatnagar, Q. M. J. Wu, and Z. Liu, "Directive contrast based multimodal medical image fusion in NSCT domain," *IEEE Trans. Multimedia*, vol. 15, no. 5, pp. 1014–1024, Aug. 2013.
- [20] F. E. Ali, I. M. El-Dokany, A. A. Saad, and F. A. El-Samie, "A curvelet transform approach for the fusion of MR and CT images," *J. Mod. Opt.*, vol. 57, no. 4, pp. 273–286, Jan. 2010.
- [21] S. Li, X. Kang, L. Fang, J. Hu, and H. Yin, "Pixel-level image fusion: A survey of the state of the art," *Inf. Fusion*, vol. 33, pp. 100–112, Jun. 2017.
- [22] M. Kumar and S. Dass, "A total variation-based algorithm for pixellevel image fusion," *IEEE Trans. Image Process.*, vol. 18, no. 9, pp. 2137–2143, Sep. 2009.
- [23] H. Li, Z. Yu, and C. Mao, "Fractional differential and variational method for image fusion and super-resolution," *Neurocomputing*, vol. 171, pp. 138–148, Jan. 2016.
- [24] Y. Liu, S. Liu, and Z. Wang, "A general framework for image fusion based on multi-scale transform and sparse representation," *Inf. Fusion*, vol. 24, pp. 147–164, Jul. 2015.
- [25] A. Loza, D. Bull, N. Canagarajah, and A. Achim, "Non-Gaussian modelbased fusion of noisy images in the wavelet domain," *Comput. Vis. Image Understand.*, vol. 114, no. 1, pp. 54–65, 2010.
- [26] P. A. Hagargi and D. Shubhangi, "Brain tumor MR image fusion using most dominant features extraction from wavelet and curvelet transforms," *Brain*, vol. 5, no. 5, pp. 33–38, 2018.

- [27] S. Paris, S. W. Hasinoff, and J. Kautz, "Local Laplacian filters: Edge-aware image processing with a Laplacian pyramid," *Commun. ACM*, vol. 58, no. 3, pp. 81–91, Feb. 2015.
- [28] E. Candès, L. Demanet, D. Donoho, and X. Ying, "Fast discrete curvelet transforms," *Multiscale Model. Simulation*, vol. 5, no. 3, pp. 861–899, Sep. 2006.
- [29] G. Kutyniok and D. Labate, "Introduction to shearlets," in *Shearlets: Multiscale Analysis for Multivariate Data*. Cambridge, MA, USA: Birkhäuser, 2012, pp. 1–38.
- [30] D. D.-Y. Po and M. N. Do, "Directional multiscale modeling of images using the contourlet transform," *IEEE Trans. Image Process.*, vol. 15, no. 6, pp. 1610–1620, Jun. 2006.
- [31] G. Easley, D. Labate, and W.-Q. Lim, "Sparse directional image representations using the discrete shearlet transform," *Appl. Comput. Harmon. Anal.*, vol. 25, no. 1, pp. 25–46, Jul. 2008.
- [32] R. Coifman and D. L. Donoho, "Translation invariant denoising," in Wavelets and Statistics (Lecture Notes in Statistics), vol. 103. New York, NY, USA: Springer-Verlag, May 1995, pp. 125–150.
- [33] W.-Q. Lim, "The discrete shearlet transform: A new directional transform and compactly supported shearlet frames," *IEEE Trans. Image Process.*, vol. 19, no. 5, pp. 1166–1180, May 2010.
- [34] X. Liu, W. Mei, and H. Du, "Multi-modality medical image fusion based on image decomposition framework and nonsubsampled shearlet transform," *Biomed. Signal Process. Control*, vol. 40, pp. 343–350, Feb. 2018.
- [35] J. M. Fadili and L. Boubchir, "Analytical form for a Bayesian wavelet estimator of images using the Bessel K form densities," *IEEE Trans. Image Process.*, vol. 14, no. 2, pp. 231–240, Feb. 2005.
- [36] E. Jabason, M. O. Ahmad, and M. S. Swamy, "Statistical modeling of multimodal neuroimaging data in non-subsampled shearlet domain using the student's t location-scale distribution," in *Proc. IEEE Int. Symp. Circuits Syst. (ISCAS)*, May 2017, pp. 1–4.
- [37] R. K. Kwan, A. C. Evans, and G. B. Pike, "MRI simulation-based evaluation of image-processing and classification methods," *IEEE Trans. Med. Imag.*, vol. 18, no. 11, pp. 1085–1097, Nov. 1999. [Online]. Available: http://brainweb.bic.mni.mcgill.ca/brainweb/
- [38] C. R. Jack et al., "The Alzheimer's disease neuroimaging initiative (ADNI): MRI methods," J. Magn. Reson. Imag., vol. 27, no. 4, pp. 685–691, Apr. 2008. [Online]. Available: http://adni.loni.usc.edu/
- [39] K. A. Johnson and J. A. Becker. (1999). *The Whole Brain Atlas*. Accessed: Jun. 2018. [Online]. Available: http://www.med.harvard.edu/aanlib/
- [40] K. Guo, D. Labate, W.-Q. Lim, G. Weiss, and E. Wilson, "Wavelets with composite dilations and their MRA properties," *Appl. Comput. Harmon. Anal.*, vol. 20, no. 2, pp. 202–236, 2006.
- [41] K. Guo, D. Labate, and W. Lim, "Edge analysis and identification using the continuous shearlet transform," *Appl. Comput. Harmon. Anal.*, vol. 27, no. 1, pp. 24–46, 2009.
- [42] D. Labate, W.-Q. Lim, G. Kutyniok, and G. Weiss, "Sparse multidimensional representation using shearlets," *Proc. SPIE*, vol. 5914, Aug. 2005, Art. no. 59140U.
- [43] M. Xu, Image Registration and Image Fusion: Algorithms and Performance Bounds. Syracuse, New York, NY, USA: Syracuse Univ., 2009.
- [44] L. Sendur and I. W. Selesnick, "Bivariate shrinkage with local variance estimation," *IEEE Signal Process. Lett.*, vol. 9, no. 12, pp. 438–441, Dec. 2002.
- [45] S. Jackman, Bayesian Analysis for the Social Sciences. Hoboken, NJ, USA: Wiley, 2009.
- [46] K. L. Lange, R. J. A. Little, and J. M. G. Taylor, "Robust statistical modeling using the t distribution," *J. Amer. Stat. Assoc.*, vol. 84, no. 408, pp. 881–896, 1989.
- [47] D. M. Endres and J. E. Schindelin, "A new metric for probability distributions," *IEEE Trans. Inf. Theory*, vol. 49, no. 7, pp. 1858–1860, Jul. 2003.
- [48] T. Stathaki, Image Fusion: Algorithms and Applications. New York, NY, USA: Academic Press, 2011.
- [49] J. Zhao, R. Laganiere, and Z. Liu, "Performance assessment of combinative pixel-level image fusion based on an absolute feature measurement," *Int. J. Innov. Comput. Inf. Control*, vol. 3, no. 6, pp. 1433–1447, 2007.
- [50] Z. Wang, A. C. Bovik, H. R. Sheikh, and E. P. Simoncelli, "Image quality assessment: From error visibility to structural similarity," *IEEE Trans. Image Process.*, vol. 13, no. 4, pp. 600–612, Apr. 2004.
- [51] R. C. Gonzalez and R. E. Woods, *Digital Image Processing*, 3rd ed. Upper Saddle River, NJ, USA: Prentice-Hall, 2006.
- [52] C. S. Xydeas and V. Petrović, "Objective image fusion performance measure," *Electron. Lett.*, vol. 36, no. 4, pp. 308–309, 2000.



EMIMAL JABASON received the B.E. degree in electronics and communication engineering and the M.E. degree in communication systems from Anna University, India, in 2010 and 2012, respectively. She is currently pursuing the Ph.D. degree in electrical and computer engineering with Concordia University, Montreal, QC, Canada, under supervision of Dr. M. O. Ahmad and Dr. M.N.S. Swamy. In 2013, she joined the Department of Electronics and Communication Engineering, Er.

Perumal Manimekalai College of Engineering, Tamil Nadu, India, as an Assistant Professor. Her current research interests include biomedical image processing, statistical modeling, computer vision, and deep learning. She is a member of the IEEE Circuits and Systems Society and the IEEE Women in Engineering. She was a recipient of numerous awards, honors, and travel supports. She has served as a Reviewer for several journals and major conferences.



M. OMAIR AHMAD (S'69–M'78–SM'83–F'01) received the B.Eng. degree in electrical engineering from Sir George Williams University, Montreal, QC, Canada, and the Ph.D. degree in electrical engineering from Concordia University, Montreal. From 1978 to 1979, he was a Faculty Member with the New York University College, Buffalo, NY, USA. In 1979, he joined the faculty of Concordia University as an Assistant Professor of computer science. Subsequently, he joined the

Department of Electrical and Computer Engineering, Concordia University, where he was the Chair of the Department, from 2002 to 2005, and is currently a Professor. He was a Founding Researcher of Micronet, a Canadian Network of Centers of Excellence, from 1990 to 2004. He was a Guest Professor with Southeast University, Nanjing, China. He is also the Concordia University Research Chair (Tier I) in multimedia signal processing. He has authored in the area of signal processing and holds four patents. His current research interests include image and speech processing, biomedical signal processing, watermarking, biometrics, video signal processing and object detection and tracking, deep learning techniques in signal processing, and fast signal transforms and algorithms. In 1988, he was a member of the Admission and Advancement Committee of the IEEE. He was a recipient of numerous honors and awards, including the Wighton Fellowship from the Sandford Fleming Foundation, an induction to Provosts Circle of Distinction for Career Achievements, and the Award of Excellence in Doctoral Supervision from the Faculty of Engineering and Computer Science, Concordia University. He was the Local Arrangements Chairman of the 1984 IEEE International Symposium on Circuits and Systems. He has served as the Program Co-Chair for the 1995 IEEE International Conference on Neural Networks and Signal Processing, the 2003 IEEE International Conference on Neural Networks and Signal Processing, and the 2004 IEEE International Midwest Symposium on Circuits and Systems. He was the General Co-Chair of the 2008 IEEE International Conference on Neural Networks and Signal Processing. He is the Chair of the Montreal Chapter IEEE Circuits and Systems Society. He was an Associate Editor of the IEEE TRANSACTIONS ON CIRCUITS AND SYSTEMS PART I: FUNDAMENTAL THEORY AND APPLICATIONS, from 1999 to 2001.



M. N. S. SWAMY (S'59–M'62–SM'74–F'80) received the B.Sc. degree (Hons.) in mathematics from the University of Mysore, Mysore, India, in 1954, the Diploma degree in electrical communication engineering from the Indian Institute of Science, Bengaluru, India, in 1957, and the M.Sc. and Ph.D. degrees in electrical engineering from the University of Saskatchewan, Saskatoon, SK, Canada, in 1960 and 1963, respectively. He was conferred with the title of Honorary Professor by

the National Chiao Tong University, Hsinchu, Taiwan, in 2009. He is currently a Research Professor with the Department of Electrical and Computer Engineering, Concordia University, Montreal, QC, Canada, where he served as the Founding Chair of the Department of Electrical Engineering, from 1970 to 1977, and the Dean of engineering and computer science, from 1977 to 1993. During that time, he developed the faculty into a researchoriented one, from what was primarily an undergraduate faculty. Since 2001, he has been the Concordia Chair (Tier I) in signal processing. He has also taught at the Department of Electrical Engineering, Technical University of Nova Scotia, Halifax, NS, Canada, the University of Calgary, Calgary, AB, Canada, and the Department of Mathematics, University of Saskatchewan. He has published in the areas of number theory, circuits, systems, and signal processing and holds five patents. He has coauthored nine books and five book chapters. He was a Founding Member of Micronet, a Canadian Network of Centers of Excellence, from 1990 to 2004, and also its Coordinator of Concordia University. He is a Fellow of the Institute of Electrical Engineers, U.K., the Engineering Institute of Canada, the Institution of Engineers, India, and the Institution of Electronic and Telecommunication Engineers, India. He was inducted to the Provosts Circle of Distinction for career achievements, in 2009. He was a recipient of many IEEE-CAS Society awards, including the Education Award, in 2000, the Golden Jubilee Medal, in 2000, and the 1986 Guillemin-Cauer Best Paper Award. He has served as the Program Chair for the 1973 IEEE Circuits and Systems (CAS) Symposium, the General Chair for the 1984 IEEE CAS Symposium, the Vice Chair of the 1999 IEEE CAS Symposium, and a member of the Board of Governors of the CAS Society. He has been the Editor-in-Chief of the journal Circuits, Systems and Signal Processing (CSSP), since 1999. Recently, CSSP has instituted the Best Paper Award in his name. He has served as the Editor-in-Chief for the IEEE TRANSACTIONS ON CIRCUITS AND SYSTEMS I, from 1999 to 2001, and an Associate Editor for the IEEE TRANSACTIONS ON CIRCUITS AND SYSTEMS, from 1985 to 1987. He has served the IEEE in various capacities, such as the President Elect, in 2003, the President, in 2004, the Past-President, in 2005, the Vice President (publications), from 2001 to 2002, and the Vice President, in 1976.